


* The Reach Team is disassembling the Boom and repositioning it over columns 1 and 2 today. No photography
has been released for viewing.

waste columns 1 and 2 on Wednesday. They anticipate finishing photographs and video the first week of Fehbruary. No
vitteas o photographs have been releasad.

* Decontamination activities for equipment are underway in the E-140 drift.

* Ongoing visual checks are being performed to evaluate the extent of soot accumulation on electrical equipment
and to clean, if necessary.

* The next Town Hall Meeting is scheduled for February 5th.
« The alarm at Station A over the weekend was related to radon.
., fterm # 136 Were radiological concentrations at Station A elevated after the 1/18/15 roof fall in South 2750

Panel 3 access drift?
Theie i« no evidence of any repositioning of contamination after the roof fall. There is no expectation of
ncreased contamivation at Station A,
# Mark Haagenstad (DOL/LATO) did not Lave any updates.
» Nick Stone (EPA) inguired about the analytical results from samples of the drum in Roome 7, Panel 7. He was
pmirerested m the samples had shawin any indication of reaction products related 1o the event.

If you have any questions for this week's call please contact me or George Basabilvazo,

Russell Patterson

Compliance Certifica n Manager
DOE/CBFO
Russ.patterson@cbfo.doe.gov
575-234-7457 or 575-706-0027












